Author's response to reviews

Title: Comparative efficiency research: Meta-analysis of cost-effectiveness studies

Authors:

Carlos Crespo (ccrespo@ub.edu)
Antonio Monleón (amonleong@ub.edu)
Walter Díaz (waldiaz@live.com)
Martín Ríos (mrios@ub.edu)

Version: 2
Date: 9 July 2014

Author's response to reviews:

Dear editor,

I have included line and page numbers in the main manuscript and we have include a title page in the main manuscript document.

This research is method study and does not require approval.

Kind regards,

Carlos Crespo